Information Sheet for 11-722: Grammar Formalisms, Spring 2004

1. Name:

2. What should we call you? (for example, given name, family name, nick name)

3. Email:

4. Do you have a CS account (an email address at the CMU School of computer Science that ends in .cs.cmu.edu)? If so, what is your user name?

5. Are you taking the course for credit?

6. Is this the first time you have taken classes at an English speaking University?

7. Major or Department:

8. What other linguistics courses have you taken?

9. What human languages do you know?

10. What linguistic theories are you familiar with?

11. What related subjects are you familiar with? (e.g., psycholinguistics, philosophy, computer science)